Author's response to reviews

Title: The role of mother-in-law in the prevention of mother-to-child transmission of HIV, northern Tanzania

Authors:

Eli Fjeld Falnes (eli.fjeld@cih.uib.no)  
Karen Marie Moland (Karen.Marie.Moland@hib.no)  
Thorkild Tylleskær (thorkild.tylleskar@cih.uib.no)  
Marina Manuela de Paoli (marina.de.paoli@fafo.no)  
Sebalda Charles Leshabari (seolesh@yahoo.com)  
Ingunn MS Engebretsen (ingunn.engebretsen@cih.uib.no)

Version: 3 Date: 1 July 2011

Author's response to reviews: see over
The Editor,
BMC Public Health

Dear Editor,

RE: RE-SUBMISSION OF MANUSCRIPT ENTITLED: THE ROLE OF MOTHER-IN-LAW IN THE PREVENTION OF MOTHER-TO-CHILD TRANSMISSION OF HIV, NORTHERN TANZANIA

We are pleased to re-submit a revised version of our manuscript attached here by Eli Fjeld Falnes, Karen Marie Moland, Thorkild Tylleskär, Marina Manuela de Paoli, Sebalda Charles Leshabari and Ingunn M.S. Engebretsen. It is entitled “The role of mother-in-law in the prevention of mother-to-child transmission of HIV, northern Tanzania”. It is submitted for your consideration and possible publication in the BMC Public Health. It is not and will not be published elsewhere while under your consideration.

We have revised the manuscript according to revisions by reviewer 2 and the Editorial Production Team.

Correspondence can be addressed to Eli Fjeld Falnes.

We are looking forward to hearing from you.

Eli Fjeld Falnes
MD, PhD candidate
Centre for International Health
Årstadveien 215th floor
N-5009 Bergen, Norway
Mobile: +47 95 20 67 01
Email: [Eli.Fjeld@cih.uib.no]

Ingunn M.S. Engebretsen
MD, PhD
Centre for International Health
Årstadveien 215th floor
N-5009 Bergen, Norway
Mobile: +47 48 07 44 10